Author’s response to reviews

Title: Programme Reporting Standards (PRS) for improving the reporting of sexual, reproductive, maternal, newborn, child and adolescent health programmes

Authors:

Anna Kågesten (akaagesten@jhu.edu;anna.kagesten@gmail.com)
Özge Tunçalp (tuncalpo@who.int)
Anayda Portela (portelaa@who.int)
Moazzam Ali (alimoa@who.int)
Nhan Tran (trann@who.int)
Metin Gülmezoglu (gulmezoglum@who.int)

Version: 1 Date: 22 May 2017

Author’s response to reviews:

Dear Editors-in-Chief,

We are pleased to hear that our manuscript entitled Programme Reporting Standards for improving the reporting of sexual, reproductive, maternal, newborn, child and adolescent health programmes was accepted with minor revisions for publication in BMC Medical Research Methodology. We thank the Editor and the reviewers for their thoughtful and extensive feedback, which allowed us to further improve the manuscript.

Based on the reviewer comments we have made extensive revisions to the manuscript; all details have been noted in the attached response to reviewers. Of special importance is that we have elaborated on the Delphi survey process and included a table to present the results of the ranking across all 3 rounds.

We greatly appreciate the care given to this manuscript. We hope that with the final revisions that we have made it will be found to be acceptable for publication. Again, we certify that the authors each contributed to and approved of the design, analyses and writing of the manuscript.
and that it is not under consideration in any other journal. We look forward to a favorable response.

Sincerely,

Özge Tunçalp (on behalf of the co-authors)

WHO Department of Reproductive Health and Research